Author's response to reviews

Title: The effect of five years versus two years of specialised assertive intervention for first episode psychosis: the randomised OPUS II clinical trial

Authors:

Marianne Melau (marianne.melau@regionh.dk)
Pia Jeppesen (pia.jep@dadlnet.dk)
Anne Thorup (aty@dadlnet.dk)
Mette Bertelsen (mbe_76@yahoo.dk)
Lone Petersen (lonepetersen@comxnet.dk)
Christian Gluud (cgluud@ctu.rh.dk)
Gertrud Krarup (gerkrar@rm.dk)
Merete Nordentoft (merete.nordentoft@dadlnet.dk)

Version: 2 Date: 8 February 2011

Author's response to reviews:

Dear Jessica White,

Thank you for your comments.

I have now modified the manuscript according to the list of formatting changes you mailed me.

Following changes have been made:

• A discussions section added to the abstract
• A discussions section within the main manuscript
• A list of abbreviations used in the manuscript and their meanings (in the main manuscript after the discussions section)
• An acknowledgement and funding section in the main manuscript
• A figure title and legend section after the reference list

Further more I add copies of the original funding and ethics documentation (4 files) in "additional material files".

Best regards

Marianne Melau

M.Sc., research assistant, ph.d. student
Research Unit
Psychiatric Centre Copenhagen
Bispebjerg Bakke 23
DK - 2400 NV
Phone: +45 3531 6590 or +45 2060 7551
mail:marianne.melau@regionh.dk